Volumes of quasifuchsian manifolds

Jean-Marc Schlenker

ABSTRACT. Quasifuchsian hyperbolic manifolds, or more generally convex co-compact hyperbolic manifolds, have infinite volume, but they have a well-defined "renormalized" volume. We outline some relations between this renormalized volume and the volume, or more precisely the "dual volume", of the convex core. On one hand, there are striking similarities between them, for instance in their variational formulas. On the other, objects related to them tend to be within a bounded distance. These analogies and proximities lead to several questions. Both the renormalized volume and the dual volume can be used for instance to bound the volume of the convex core in terms of the Weil-Petersson distance between the conformal metrics at infinity.

CONTENTS

1.	Two relations between surfaces and quasifuchsian manifolds	321	
1.1.	The Teichmüller and Fricke-Klein spaces of a surface	321	
1.2.	3-dimensional hyperbolic structures	322	
1.3.	The convex core of quasifuchsian manifolds	322	
1.4.	.4. The measured bending lamination of the boundary of the		
	convex core	323	
1.5.	Volumes of quasifuchsian manifolds	323	
1.6.	The dual Bonahon-Schläfli formula	324	
1.7. The holomorphic quadratic differential at infinity		325	
1.8.	A first variational formula for the renormalized volume	325	
1.9.	The measured foliation at infinity and Schläfli formula at		
	infinity	325	
1.10	. The Schläfli formula for the renormalized volume	325	
1.11	. Comparing and relating the two viewpoints	326	
Outline of the content			

Partially supported by UL IRP grant NeoGeo and FNR grants INTER/ANR/15/11211745 and OPEN/16/11405402. The author also acknowledge support from U.S. National Science Foundation grants DMS-1107452, 1107263, 1107367 "RNMS: GEometric structures And Representation varieties" (the GEAR Network).

Ack	nowledgement	327	
2.	Background material	327	
2.1.	The Fischer-Tromba metric	327	
2.2.	Complex projective structures on a surface	328	
2.3.	The Schwarzian derivative	328	
2.4.	The measured bending lamination on $\partial C(M)$	329	
2.5.	The grafting map	330	
2.6.	The energy of harmonic maps and the Gardiner formula	330	
2.7.	Extremal lengths of measured foliations	331	
2.8.	Quasifuchsian manifolds	332	
3.	The renormalized volume of quasifuchsian manifolds	332	
3.1.	Outline	332	
3.2.	Equidistant foliations near infinity	333	
3.3.	Definition and first variation of the W-volume	334	
3.4.	First variation of the W-volume from infinity	336	
3.5.	Definition of the renormalized volume	336	
3.6.	The variational formula (3)	337	
3.7.	Further properties	338	
4.	The extremal length and the measured foliation at infinity	338	
4.1.	The measured foliation at infinity	338	
4.2.	Proof of Theorem 1.7	339	
5.	Comparisons	340	
5.1.	Outline	340	
5.2.	Comparing the renormalized volume and the dual volume	340	
5.3.	An upper bound on the renormalized volume	342	
5.4.	An upper bound on the dual volume	343	
5.5.	Estimates from the dual volume	343	
6.	Applications	344	
6.1.	Bounding the volume of the convex core using the renormalized	ł	
	volume or the dual volume	344	
6.2.	The volume of hyperbolic manifolds fibering over the circle	344	
6.3.	Systoles of the Weil-Petersson metric on moduli space	345	
6.4.	Entropy and hyperbolic volume of mapping tori	345	
7.	Questions and perspectives	346	
7.1.	Convexity of renormalized volume	346	
7.2.	The measured foliation at infinity	347	
7.3.	Comparing the foliation at infinity to the measured bending		
	lamination	348	
7.4.	Extension to convex co-compact or geometrically finite		
	hyperbolic manifolds	348	
7.5.	Higher dimensions	349	
Refe	References		

1. Two relations between surfaces and quasifuchsian manifolds

1.1. The Teichmüller and Fricke-Klein spaces of a surface. Consider a closed oriented surface S of genus at least 2, and let $M = S \times \mathbb{R}$. One can then define \mathcal{T}_S , the Teichmüller space of S, as the space of complex structures on S considered up to diffeomorphisms isotopic to the identity. It is also interesting to introduce \mathcal{F}_S , the Fricke space of S, defined as the space of hyperbolic structures on S considered up to diffeomorphisms isotopic to the identity.

The Poincaré-Riemann uniformization theorem provides a diffeomorphism \mathcal{P}_S between \mathcal{T}_S and \mathcal{F}_S , but keeping different notations might be preferable here. The geometry of these spaces develops along related but distinct lines. For instance, the cotangent space $T_c^*\mathcal{T}_S$ at a complex structure $c \in \mathcal{T}_S$ is classically identified with the space of holomorphic quadratic differentials on (S, c) and the tangent bundle $T_c\mathcal{T}_S$ with the space of harmonic Beltrami differentials on (S, c) (see e.g. [1]), while the cotangent space $T_h^*\mathcal{F}_S$ at a hyperbolic metric h can be identified with the space of measured geodesic laminations on (S, h), and the tangent space $T_h\mathcal{F}_S$ is identified with the space of traceless Codazzi 2-tensors on (S, h), see Section 2.1.

The Teichmüller space \mathcal{T}_S of S can be equipped with a Riemannian metric, the Weil-Petersson metric g_{WP} . It is simpler to define it on the cotangent space. Given two holomorphic quadratic differentials $q, q' \in T_c^* \mathcal{T}_S$ at a complex structure c, their scalar product is defined as:

$$g_{WP}(q,q') = \int_S \frac{q\overline{q'}}{h} \, ,$$

where $h = \mathcal{P}_S(c)$ is the hyperbolic metric uniformizing c. Here the quotient $q\overline{q'}/h$ makes sense as an area form on S, as can be seen using a local coordinate z: if $q = fdz^2$ and $q' = f'dz^2$, and if $h = \rho |dz|^2$, then $q\overline{q'}/h = (f\overline{f'}/\rho)|dz|^2$. This scalar product on cotangent vectors defines an identification between the cotangent space $T_c^*\mathcal{T}_S$ and the tangent space $T_c\mathcal{T}_S$, a scalar product on the tangent space $T_c\mathcal{T}_S$, and therefore a Riemannian metric on \mathcal{T}_S . The Weil-Petersson Riemannian metric is known to be Kähler [76, 2] and to have negative sectional curvature [3]. It is incomplete, but geodesically convex [81, 82].

On the side of the Fricke space, a closely related analog of the Weil-Petersson metric was defined by Fischer and Tromba [28]. Let $h \in \mathcal{F}_S$ be a hyperbolic metric, and let $[k_1], [k_2] \in T_h \mathcal{F}_S$ be the tangent vector fields defined by two traceless, Codazzi symmetric 2-tensors on (S, h). The Fischer-Tromba metric is defined as:

$$g_{FT}([k_1], [k_2]) = \frac{1}{8} \int_S \langle k_1, k_2 \rangle_h da_h$$

This metric then corresponds to the Weil-Petersson metric, see [28]:

$$g_{WP} = \mathcal{P}_S^* g_{FT}$$
.

1.2. 3-dimensional hyperbolic structures. There are deep relations between the geometry of \mathcal{T}_S (resp. \mathcal{F}_S) and hyperbolic structures on 3dimensional manifolds. These relations develop differently for \mathcal{T}_S and for \mathcal{F}_S and remain partly conjectural. Our main goal here is to describe the analogies between these relations. We focus here on quasifuchsian manifolds, and will only briefly mention the extension to convex co-compact hyperbolic manifolds. To simplify notations, we set $M = S \times \mathbb{R}$. Then the boundary ∂M of (M,g) can be identified canonically to $S \cup \overline{S}$, where \overline{S} is S with the opposite orientation, so that $\mathcal{T}_{\partial M} = \mathcal{T}_S \times \mathcal{T}_{\overline{S}}$. Both \mathcal{T}_S and $\mathcal{T}_{\overline{S}}$ can be identified with the space of conformal metrics on S, and we will often identify \mathcal{T}_S with $\mathcal{T}_{\overline{S}}$ in this manner.

DEFINITION 1.1. A quasifuchsian structure on M is a complete hyperbolic metric g on M such that (M, g) contains a non-empty compact geodesically convex subset. We denote by \mathcal{QF}_S the space of quasifuchsian hyperbolic structures on M, considered up to isotopies.

The relation between quasifuchsian hyperbolic manifolds and the Teichmüller space of S rests on the Bers Double Uniformization Theorem, see [6].

THEOREM 1.2 (Bers). Given a quasifuction structure $g \in Q\mathcal{F}_S$, the asymptotic boundary $\partial_{\infty}M$ of (M,g) is equipped with a complex structure $c = (c_+, c_-)$, and each such $c \in \mathcal{T}_{\partial M}$ is obtained from a unique $g \in Q\mathcal{F}_S$.

We are also interested in the relation between quasifuchsian manifolds and the Fricke space \mathcal{F}_S . This relation can be understood through a *conjectural* statement, due to Thurston, which is analogous to the Bers Double Uniformization Theorem.

1.3. The convex core of quasifuchsian manifolds. By definition, a quasifuchsian hyperbolic manifold contains a non-empty, compact, geodesically convex subset. Since the intersection of two non-empty geodesically convex subsets is geodesically convex, any quasifuchsian manifold (M, g) contains a unique smallest non-empty geodesically convex subset, which is compact. It is called the *convex core* of (M, g), and will be denoted here by C(M).

There is a rather special case where C(M) is a totally geodesic surface in (M,g) — in that case, (M,g) is a *Fuchsian* manifold. In all other cases, C(M) has non-empty interior, and its boundary is the disjoint union of two surfaces homeomorphic to S, denoted here by $\partial_+ C(M)$ and $\partial_- C(M)$.

Thurston [71] noted that since C(M) is a minimal convex set, its boundary has no extreme point, so $\partial_+C(M)$ and $\partial_-C(M)$ are convex *pleated* surfaces. Their induced metrics are hyperbolic (i.e. of constant curvature -1), and this defines two points $m_+, m_- \in \mathcal{F}_S$.

CONJECTURE 1.3 (Thurston). For all $(m_+, m_-) \in \mathcal{F}_S \times \mathcal{F}_S$, there exists a unique $g \in \mathcal{QF}_S$ such that the induced metrics on $\partial_+C(M)$ and $\partial_-C(M)$ are m_+ and m_- , respectively. The existence part of this statement is known since work of Labourie [48], Epstein and Marden [25] and Sullivan [67].

Conjecture 1.3 is of course analogous to the Bers Simultaneous Uniformization Theorem, when one replaces the complex structure (or conformal metric) at infinity by the induced metric on the boundary of the convex core. One main goal here is to extend this analogy. The other goal is to extend the *comparisons* between objects associated to the Teichmüller theory of S, read at infinity, and objects associated to the Fricke space of S, read from the boundary of the convex core. For the conformal metric at infinity and induced metric on the boundary of the convex core, the following result provides a bound on the distance between the two.

THEOREM 1.4 (Sullivan, Epstein-Marden). There exists a universal constant K such that m_{\pm} are K-quasiconformal to c_{\pm} , respectively.

The constant K was long conjectured to be equal to 2, but is actually larger than 2.1, see [26].

1.4. The measured bending lamination of the boundary of the convex core. To understand the definition of "dual volume" that plays a central role below, we need another important notion: the bending measured lamination on the boundary of the convex core. This is the quantity that records in what manner the boundary of the convex core is "pleated" in M. It is a transverse measure on a geodesic lamination on $\partial C(M)$. A short description of some of its main properties, and of the main properties of measured laminations more generally, can be found in Section 2.4.

1.5. Volumes of quasifuchsian manifolds. Quasifuchsian hyperbolic manifolds have infinite volume. However, techniques originating from physics make it possible to define a *renormalized* volume, see Section 3. This renormalized volume is closely related to the Liouville functional, see e.g. [69, 68, 70, 44]. It determines a function $V_R : \mathcal{QF}_S \to \mathbb{R}$ which can also be considered, through the Bers Simultaneous Uniformization Theorem, as a function $V_R : \mathcal{T}_S \times \mathcal{T}_{\overline{S}} \to \mathbb{R}$. When $c_- \in \mathcal{T}_{\overline{S}}$ is fixed, the function $V_R(\cdot, c_-) : \mathcal{T}_S \to \mathbb{R}$ is a Kähler potential for the Weil-Petersson metric on \mathcal{T}_S , a fact that we will not develop here. (A proof can be found in [45, Section 9].)

The convex core C(M), on the other hand, has a well-defined volume, and this defines a function $V_C : \mathcal{QF}_S \to \mathbb{R}_{>0}$, the volume of the convex core. It should be clear from the considerations that follow, however, that V_C is not the "right" analog of the renormalized volume, and we rather consider the *dual volume*.

DEFINITION 1.5. The *dual volume* of the convex core of a quasifuchsian manifold (M, g) is

$$V_C^*(M) = V_C(M) - \frac{1}{2}L_m(l)$$

where m is the induced metric on the boundary of the convex core, and l is it's measured bending lamination.

The dual volume can be defined for a more general geodesically convex subset $K \subset M$. For a convex subset with smooth boundary, it is defined as

$$V^*(K) = V(K) - \frac{1}{2} \int_{\partial K} H da ,$$

where H is the mean curvature of ∂K (defined as the sum of its principal curvatures) and da is the area form of the induced metric on the boundary of K.

The reason for the term "dual volume" is that, if P is a convex polyhedron in \mathbb{H}^3 and its "dual volume" V^* is defined in the same manner as $V^* = V - \sum_e L_e \theta_e$, where the sum is over the edges and L_e (resp. θ_e) is the length (resp. exterior dihedral angle) of edge e, then V^* is equal to the volume, suitably defined, of the dual polyhedron in the de Sitter space, see [36]. For quasifuchsian manifolds, a similar interpretation is possible, but only in a *relative* manner. A quasifuchsian manifold has a de Sitter counterpart M^* , which is a pair of globally hyperbolic de Sitter manifolds M^*_+ , M^*_- , see [54, 5]. Any convex compact subset $K \subset M$ has a pair of dual convex subsets $K^*_+ \subset M^*_+$, $K^*_- \subset M^*_-$. If K and \bar{K} are two subsets of M with $K \subset \bar{K}$, then $\bar{K}^* \subset K^*$, and $V^*(\bar{K}) - V^*(K) = V(K^* \setminus \bar{K}^*)$. We do not delve more onto this topic and refer the interested reader to [51] for details.

1.6. The dual Bonahon-Schläfli formula. The classical Schläfli formula [55] expresses the first-order variation of the volume of a hyperbolic polyhedron $P \subset \mathbb{H}^3$ in terms of the variation of its exterior dihedral angles as follows:

$$\dot{P} = \frac{1}{2} \sum_{e} l(e) \dot{\theta}(e) ,$$

where the sum is over the edges of P, l(e) is the length and $\theta(e)$ the exterior dihedral angle of edge e.

Bonahon [9, 8] extended this classical formula to the convex cores of quasifuchsian (or more generally convex co-compact) hyperbolic manifolds. In a first-order deformation of a quasifuchsian manifold (M, g), corresponding say to a first-order variation of the holonomy representation,

(1)
$$\dot{V} = \frac{1}{2}L_m(\dot{l}) \; .$$

Bonahon showed that \dot{l} , the first-order variation of l, makes senses as a *Hölder cocycle*, and has a well-defined length, so that (1) makes sense.

The dual Bonahon-Schläfli formula is the analog of the Bonahon-Schläfli variational formula for the dual volume (see [46]). It is a direct consequence of (1):

(2)
$$\dot{V}_C^* = -\frac{1}{2}(dL(l))(\dot{m})$$
.

Note however that the interpretation of (2) is much simpler than that of (1), since now the right-hand term is simply the differential of an analytic

function — the length of l — applied to a tangent vector to $\mathcal{F}_{\partial C(M)}$. We will see below that Equation (2) is closely analogous to the variational formula for the renormalized volume. Before stating this formula, we need to better understand the geometric data at infinity of quasifuchsian manifolds.

1.7. The holomorphic quadratic differential at infinity. We now introduce what we believe to be a natural analog at infinity of the measured lamination on the boundary of the convex core. This is a measured *lamination*, defined as follows. Given a quasifuchsian structure $g \in Q\mathcal{F}_S$ on M, we have seen that the asymptotic boundary $\partial_{\infty}M$ is the disjoint union of two disjoint Riemann surfaces (S, c_+) and (\bar{S}, c_-) . In fact, each of those surfaces is equiped not only with a complex structure c_{\pm} , but also with a *complex projective structure* σ_{\pm} , see Section 2.8.

The Schwarzian derivative (see Section 2.3) provides the tool to compare σ_{\pm} to $\sigma_F(c_{\pm})$, the Fuchsian complex projective structure associated to c_{\pm} . This yields a holomorphic quadratic differential q_{\pm} on (S, c_{+}) and (\bar{S}, c_{-}) , or in other terms a holomorphic quadratic differential q on $\partial_{\infty} M$, which we call the holomorphic quadratic differential at infinity.

1.8. A first variational formula for the renormalized volume. The renormalized volume also satisfies a simple variational formula, see Section 3.6.

(3)
$$\dot{V}_R = \operatorname{Re}(\langle q, \dot{c} \rangle)$$
,

where q is considered as a vector in the complexified cotangent space to \mathcal{T}_S at c, and $\langle \cdot, \cdot \rangle$ is the duality bracket.

We will see below that this first variational formula can be formulated in a way that makes it similar to (2), using the extremal length of a measured foliation at infinity instead of the hyperbolic length of a measured lamination on the boundary of the convex core.

1.9. The measured foliation at infinity and Schläfli formula at infinity. A holomorphic quadratic differential q on a Riemann surface (S, c) determines canonically two measured foliations, the *horizontal* and *vertical* foliations. The leaves of the horizontal (resp. vertical) foliation are the integral curves of the vector fields u such that $q(u, u) \in \mathbb{R}_{>0}$ (resp. $\in R_{<0}$), see [27].

DEFINITION 1.6. The measured foliation at infinity of M, denoted by $f \in \mathcal{MF}_{\partial M}$, is the horizontal foliation of the holomorphic quadratic differential q of M.

1.10. The Schläfli formula for the renormalized volume. There is a simple variational formula for the renormalized volume, in terms of q and of the variation of the conformal structure at infinity, Equation (3). Here we write this variational formula in another way, involving the measured foliation at infinity. Instead of the hyperbolic length of the measured bending lamination, as for the dual volume, this formula involves the *extremal length* of the measured foliation at infinity.

Recall that given a Riemann surface (S, c) and a simple closed curve γ on S, the extremal length $ext(\gamma)$ of γ can be defined as the supremum of the inverses of the conformal moduli of annuli embedded in S with meridian isotopic to γ .

THEOREM 1.7. In a first-order variation of M, we have

(4)
$$\dot{V}_R = -\frac{1}{2}(dext(f))(\dot{c}) \; .$$

Here $\operatorname{ext}(f)$ is considered as a function over the Teichmüller space of the boundary $\mathcal{T}_{\partial M}$. The right-hand side is the differential of this function, evaluated on the first-order variation of the complex structure on the boundary.

1.11. Comparing and relating the two viewpoints. Theorem 1.7, and the analogy between (2) and (4), suggests an analogy between the properties of quasifuchsian manifolds considered from the boundary of the convex core and from the boundary at infinity. For instance, on the boundary of the convex core, we have the following upper bound on the length of the bending lamination, see [13, Theorem 2.16].

THEOREM 1.8 (Bridgeman, Brock, Bromberg). $L_{m_{\pm}}(l_{\pm}) \leq 6\pi |\chi(S)|$.

Similarly, on the boundary at infinity, we have the following result, proved in Section 4.1.

THEOREM 1.9. $ext_{c_{\pm}}(f_{\pm}) \leq 3\pi |\chi(S)|.$

On the convex core	At infinity
Induced metric m	Conformal structure at infinity c
Thurston's conjecture on prescrib-	Bers' Simultaneous Uniformization
ing m	Theorem
Measured bending lamination l	measured foliation f
Hyperbolic length of l for m	Extremal length of f for c
Volume of the convex core V_C	Renormalized volume V_R
Dual Bonahon-Schläfli formula	Theorem 1.7
$\dot{V}_{C}^{*} = -\frac{1}{2}(dL(l))(\dot{m})$	$\dot{V}_R = -\frac{1}{2}(dext(f))(\dot{c})$
Bound on $L_m(l)$ [12, 13]	Theorem 1.9
$L_{m_{\pm}}(l_{\pm}) \le 6\pi \chi(S) $	$\operatorname{ext}_{c_{\pm}}(f_{\pm}) \le 3\pi \chi(S) $
Brock's upper bound on V_C [15]	Upper bound on V_R [64]

TABLE 1. Infinity vs the boundary of the convex core

This analogy, briefly described in Table 1, suggests a number of questions (see Section 7) since it appears that, at least up to a point, results known on the boundary of the convex core might hold also on the boundary at infinity, and conversely.

Another series of questions stems from comparing the data on the boundary of the convex core to the corresponding data on the boundary at infinity. For instance, it was proved by Sullivan that the induced metric on the boundary of the convex core is uniformly quasi-conformal to the conformal metric at infinity (see [25, 26]), and one can ask whether similar statements hold for other quantities. We do not delve much on those questions here, see Section 7.4 for a question in this direction.

Outline of the content. Section 2 contains background material on a variety of topics that are considered or used in the paper. The renormalized volume is defined in Section 3, and its main properties proved. Section 4 contains the proof of the Schläfli-type formula for the renormalized volume, (4), while section 6 explains how to obtain upper bounds on the volume of the convex core in terms of boundary data, using either the dual or the renormalized volume. It then outlines some applications, in particular results of Brock and Bromberg [16] on the systoles of the Weil-Petersson metric on moduli space and of Kojima and McShane [41] on the comparison between the entropy of a pseudo-Anosov diffeomorphism and the hyperbolic volume of its mapping torus. Finally Section 7 presents some open questions.

Acknowledgement. The author is grateful to Keaton Quinn for carefully reading a previous version of this text and noticing numerous typos and inconsistencies, and to an anonymous referee for numerous additional remarks and corrections leading to a clear improvement in the presentation.

2. Background material

This section contains a short description of some of the background material used in the paper, aiming at providing references for readers who are not familiar with certain topics.

2.1. The Fischer-Tromba metric. Let h be a hyperbolic metric on S. The tangent space $T_h \mathcal{F}_S$ to the Fricke space of S can be identified with the space of symmetric 2-tensors on S that are traceless and satisfy the Codazzi equation for h, see [28]. (In other terms, the real parts of holomorphic quadratic differentials in \mathcal{Q}_c , if c is the complex structure compatible with h on S.)

Let k, l be two such tensors and let [k], [l] be the corresponding vectors in $T_h \mathcal{F}$. Then the Weil-Petersson metric between [k] and [l] can be expressed as

$$\langle [k], [l] \rangle_{WP} = \frac{1}{8} \int_{S} \langle k, l \rangle_{h} da_{h} .$$

The right-hand side of this equation is sometimes called the Fischer-Tromba metric on \mathcal{F}_S . It is proved in [28] that this metrics corresponds to the Weil-Petersson metric on \mathcal{T}_S , through the identification of \mathcal{T}_S with \mathcal{F}_S by the Poincaré-Riemann Uniformization Theorem.

We can also relate the scalar product on symmetric 2-tensors to the natural bracket between holomorphic quadratic differentials and Beltrami differentials as follows.

LEMMA 2.1. Let X be a closed Riemann surface, and let h be the hyperbolic metric compatible with its complex structure. Let \dot{h} be a first-order deformation of h, and let μ be the corresponding Beltrami differential. Then for any holomorphic quadratic differential q on X,

$$\int_X \langle Re(q), h' \rangle_h da_h = 4Re\left(\int_X q\mu\right)$$

2.2. Complex projective structures on a surface. A complex projective structure (also called $\mathbb{C}P^1$ -structure) is a (G, X)-structure (see [71, 29]), where $X = \mathbb{C}P^1$ and $G = PSL(2, \mathbb{C})$. Such a structure can be defined by an atlas of charts with values in $\mathbb{C}P^1$, with change of coordinates in $PSL(2, \mathbb{C})$. We denote by \mathcal{CP}_S the space of $\mathbb{C}P^1$ -structures on S.

The space \mathcal{CP}_S of complex projective structures can be identified with either $T^*\mathcal{T}_S$ or $T^*\mathcal{F}_S$, itself identified with $\mathcal{F}_S \times \mathcal{ML}_S$. We describe those two identifications below. The first uses the Schwarzian derivative, while the second is through the grafting map.

2.3. The Schwarzian derivative. Let $\Omega \subset \mathbb{C}$ be an open subset, and let $f : \Omega \to \mathbb{C}$ be holomorphic. The Schwarzian derivative of f is a meromorphic quadratic differential defined as

$$\mathcal{S}(f) = \left(\left(\frac{f''}{f'} \right)' - \frac{1}{2} \left(\frac{f''}{f'} \right)^2 \right) dz^2 \; .$$

It has two remarkable properties that both follow from lengthy but direct computations based on the definition.

- (1) S(f) = 0 if and only if f is a Möbius transformation,
- (2) if $g : \Omega' \to \mathbb{C}$ is holomorphic and $f(\Omega) \subset \Omega'$ then $\mathcal{S}(g \circ f) = f^*\mathcal{S}(g) + \mathcal{S}(f)$.

It follows from those two properties that the Schwarzian derivative is defined for any holomorphic map from a surface equipped with a complex projective structure to another: given such a map, its Schwarzian derivative can be computed with respect to a coordinate chart in the domain and target surfaces, and properties (1) and (2) indicate that it actually does not depend on the choice of charts.

There are several nice geometric interpretations of the Schwarzian derivative that can be found in [72], [23] or in [21].

Given a complex structure $c \in \mathcal{T}_S$ on S, there is by the Poincaré-Riemann Uniformization Theorem a unique hyperbolic metric h_c on S compatible with c. Any hyperbolic metric has an underlying complex projective structure on S, because the hyperbolic plane can be identified with a disk in $\mathbb{C}P^1$, on which hyperbolic isometries act by elements of $PSL(2,\mathbb{C})$ fixing the boundary circle. We denote by $\sigma_F(c)$ the underlying complex projective structure of the hyperbolic metric h_c , and call it the *Fuchsian* complex projective structure of c.

Let $\sigma \in \mathcal{CP}_S$, and let $c \in \mathcal{T}_S$ be the underlying complex structure. There is a unique map $\phi : (S, \sigma) \to (S, \sigma_F(c))$ holomorphic for the underlying complex structure and isotopic to the identity. Let $q(\sigma) = \mathcal{S}(\phi)$ be its Schwarzian derivative. This construction defines a map $\mathcal{Q} : \mathcal{CP}_S \to T^*\mathcal{T}_S$, sending σ to $(c, q(\sigma))$.

The holomorphic quadratic differential $q(\sigma)$ can be considered as a cotangent vector to \mathcal{T}_S at c, so that \mathcal{Q} can be defined as a map from \mathcal{CP}_S to $T^*\mathcal{T}_S$.

The map \mathcal{Q} is known to be a homeomorphism, see [22].

2.4. The measured bending lamination on $\partial C(M)$. Although the induced metric on the boundary of the convex core is hyperbolic, the boundary surface is not (except in the Fuchsian case) totally geodesic. Rather, it is "pleated" along a locus which is a disjoint union of complete geodesics.

The simplest situation is when this pleating locus is a simple closed geodesic, or a disjoint union of such geodesics. The amount of pleating is then measured by an angle, analogous to the exterior dihedral angle at the edge of a hyperbolic polyhedron. It is quite natural then to describe the pleating as a *transverse measure* along the pleating locus: any segment transverse to the pleating locus has a weight, which is simply the sum of the pleating angles along the connected component of the pleating locus that it intersects, and this weight is constant when the segment is deformed while remaining transverse to the pleating locus. There is then a natural notion of "length" of this measured pleating lamination: it is simply the sum of products of the length of the connected component of the pleating locus by their pleating angle.

However, the pleating locus is generally much more complicated: it is a *geodesic lamination*, that is, disjoint union of simple geodesics which might be non-closed. This geodesic lamination is also equipped with a transverse measure quantifying the amount of pleating. The pleating of the surface is therefore described by a *measured geodesic lamination*.

We refer the reader to [10] for a nice introduction to geodesic laminations on hyperbolic surfaces. Here are a few key points.

- As for closed curves, the notion of measured lamination can be considered on a surface without reference to a hyperbolic metric. Given a measured lamination on S, it has a unique geodesic realization for each hyperbolic metric h on S. We will denote by \mathcal{ML}_S the space of measured laminations on S.
- \mathcal{ML}_S can be defined as the completion of the space of weighted simple closed curves (or multicurves) on S for a natural topology defined by intersection with closed curves.
- The projectivization $P\mathcal{ML}_S$ of \mathcal{ML}_S provides a compactification of \mathcal{F}_S , called the Thurston boundary, see e.g. [71, 27].

- On a hyperbolic surface (S, h), measured laminations have a welldefined length, defined by continuity from the length of weighted simple closed curves. (The length of a weighted closed curve is the product of the weight by the length of the geodesic representative of the curve.) We will denote by $L_h(l)$ the length of a measured lamination l with respect to a hyperbolic metric h on S.
- For each $l \in \mathcal{ML}_S$, the length function $L_{\cdot}(l) : \mathcal{F}_S \to \mathbb{R}_{\geq 0}$ is analytic over \mathcal{F}_S (see [40]). At each hyperbolic metric $h \in \mathcal{F}_S$, the differential of the length function, $l \mapsto d_h L_{\cdot}(l)$, provide a homeomorphism from \mathcal{ML}_S to $T_h^* \mathcal{F}_S$, so that $T^* \mathcal{F}_S$ can be identified globally with $\mathcal{F}_S \times \mathcal{ML}_S$.

2.5. The grafting map. We now turn to the description of complex projective structures on a surface in terms of hyperbolic metrics and measured laminations.

Consider first the simple situation where the measured bending lamination on $\partial_+ C(M)$ is supported on a disjoint union of simple closed curves. The upper boundary at infinity $\partial_+ M$ of M can then be decomposed as the union of two sub-domains by considering the extension to infinity of the nearest-point projection from $M \setminus C(M)$ to $\partial C(M)$. The set of points which project to the complement of the bending locus of $\partial_+ C(M)$ is projective equivalent to the complement of a lamination in $\partial_+ C(M)$ (equipped with the complex projective structure underlying its induced hyperbolic metric), while the set of points projecting to the bending lamination of $\partial_+ C(M)$ is a disjoint union of annuli, each carrying a standard complex projective structure depending on two parameters: the length and bending angle at each closed geodesic in the bending locus.

In this manner, the complex projective structure on $\partial_+ M$ can be obtained by a well-defined procedure, where $\partial_+ C(M)$ (equipped with the complex projective structure underlying its induced metric) is cut along the support of the measured lamination, and a projective annulus is inserted in each cut. Thurston called *grafting* the function sending the induced metric m and measured bending lamination l to the complex projective structure at infinity σ , and he proved that this function extends to a homeomorphism $gr: \mathcal{F}_S \times \mathcal{ML}_S \to \mathcal{CP}_S$, see [22, 38].

The grafting map therefore provides an identification of \mathcal{CP}_S and $T^*\mathcal{F}_S$, identified with $\mathcal{F}_S \times \mathcal{ML}_S$.

2.6. The energy of harmonic maps and the Gardiner formula. The proof of Equation (4) from Equation (3) uses well-known results involving the energy of harmonic maps and length of measured foliations. We recall those statements in this section and the next.

Given a measured foliation $f \in \mathcal{MF}_S$, consider its universal cover f, which is a measured foliation of \tilde{S} . One can then define the *dual tree* $T_{\tilde{f}}$ of the universal cover \tilde{f} , see e.g. [78, 56]. In the simplest case where f has

closed leaves, the vertices of the dual tree $T_{\tilde{f}}$ correspond to singular points of the foliation \tilde{f} , while each leaf of the foliation corresponds to an interior point of an edge. However for general measured foliations, $T_{\tilde{f}}$ is a *real tree*.

Let $f \in \mathcal{MF}_S$ be a measured foliation, and let T_f be its dual real tree. There is a well-defined notion of harmonic map from a Riemann surface to a real tree [79], or more generally to a non-positively curved geodesic metric space, see [32, 42, 43] or [20, Section 2.3.3] for a nice exposition. For each $c \in \mathcal{T}_S$, there is a unique equivariant harmonic map u from \tilde{S} to T_f , see [78], and it has a well-defined notion of Hopf differential, which is still a holomorphic quadratic differential on (S, c), see [66]¹ (or [20, Section 3.1.2] for an exposition). Let $E_f(c) = E(u, c)$ be its energy, and let Φ_f be its Hopf differential. The following remarkable formula can be found in [77, Theorem 1.2].

(5)
$$dE_f(\dot{c}) = -4Re(\langle \Phi_f, \dot{c} \rangle) \;.$$

Here \dot{c} is considered as a Beltrami differential, and $\langle \cdot, \cdot \rangle$ is the duality product between Beltrami differentials and holomorphic quadratic differentials.

We use below the same notations, but with S replaced by ∂M .

2.7. Extremal lengths of measured foliations. Let c be a complex structure on S, and let Q be a holomorphic quadratic differential on (S, c). Q determines two measured foliations on S, its *horizontal* and *vertical* foliations, see [37]. For any non-zero vector v tangent to a leaf of the horizontal foliation, $q(v, v) \in \mathbb{R}_{>0}$, while if v is tangent to a leaf of the vertical foliation, $q(v, v) \in \mathbb{R}_{<0}$. It is well known (see e.g. [27]) that, given a measured foliation f on (S, c), there is a unique holomorphic quadratic differential on (S, c) with horizontal measured foliation f.

Let f be a measured foliation on S and, for a given $c \in \mathcal{T}$, let Q be the holomorphic quadratic differential on S with horizontal foliation f. We will use the following relation, see [39].

LEMMA 2.2. The extremal length of f at c is the integral over S of Q,

$$ext_c(f) = \int_S |Q| \; .$$

Moreover, Wolf proved that the extremal length of a measured foliation is directly related to the energy of the harmonic map to its dual tree as follows.

THEOREM 2.3 ([80]). $Q = -\Phi_f$. Moreover,

$$E_f(c) = 2 \int_S |\Phi_f| = 2 \int_S |Q| = 2 \operatorname{ext}_c(f)$$
.

¹ref "the effect of curvature..."

2.8. Quasifuchsian manifolds. We collect here a few basic facts on quasifuchsian hyperbolic manifolds. Recall that $M = S \times \mathbb{R}$, where S is a closed oriented surface of genus at least 2.

Quasifuchsian structures on M were defined in Definition 1.1, but can also be defined as quasiconformal deformations of Fuchsian structures. Specifically, given a complete hyperbolic metric g on M, with (M, g) isometric to $\mathbb{H}^3/\rho(\pi_1 S)$, (M, g) is quasifuchsian if and only if there exists a Fuchsian representation $\rho_0: \pi_1 S \to \mathrm{PSL}(2, \mathbb{R})$ and a quasiconformal homeomorphism $\phi: \mathbb{CP}^1 \to \mathbb{CP}^1$ such that the actions of ρ_0 and ρ on \mathbb{CP}^1 are conjugated by $\phi: \rho(\gamma) = \phi^{-1} \circ \rho_0(\gamma) \circ \phi$ for any $\gamma \in \pi_1 S$.

This point of view leads to the following proposition.

PROPOSITION 2.4. Given a quasifuchsian structure $g \in \mathcal{QF}_S$ on M, (M,g) is the quotient of \mathbb{H}^3 by the image of a morphism $\rho : \pi_1 S \to \mathrm{PSL}(2,\mathbb{C})$. The corresponding action of $\pi_1 S$ on \mathbb{CP}^1 is properly discontinuous and free on each connected component of the complement of a Jordan curve Λ_ρ . Moreover, Λ_ρ is a quasicircle, that is, the image of $\mathbb{RP}^1 \subset \mathbb{CP}^1$ by a quasiconformal homeomorphism from \mathbb{CP}^1 to \mathbb{CP}^1 .

It follows that each connected component of $\mathbb{CP}^1 \setminus \Lambda_{\rho}$ corresponds to a connected component of the boundary at infinity $\partial_{\infty} M$. Since ρ acts on each by elements of $\mathrm{PSL}(2,\mathbb{C})$, each is equipped with a complex projective structure. We denote by σ the complex projective structure defined in this manner on $\partial_{\infty} M$, and by σ_{\pm} the complex projective structure defined on $\partial_{\infty,\pm} M$, the two connected components of $\partial_{\infty} M$.

3. The renormalized volume of quasifuchsian manifolds

3.1. Outline. The renormalized volume of quasifuchsian manifolds is at the intersection of two distinct developments in mathematics.

- It is closely related to the Liouville functional in complex analysis, see [69, 68, 70, 44].
- It can also be considered as the 3-dimensional case of the renormalized volume of conformally compact Einstein manifolds, see [35, 31, 30].

A definition of the renormalized volume of quasifuchsian manifolds can be found in [45, Def 8.1] or in [64, Section 3]. We recall this definition here for completeness. It is based on equidistant foliations in the neighborhood of infinity, on a notion of "W-volume" of geodesically convex subsets of a quasifuchsian manifold, and on a "renormalized" limit as $r \to \infty$ of the Wvolume of the region between the level r surfaces of a well-chosen equidistant foliation.

Another equivalent definition uses a conformal description of the metric and the behavior at 0 of a meromorphic function constructed by integration, see [33]. **3.2. Equidistant foliations near infinity.** We first define equidistant foliations in the neighborhood of infinity in a quasifuchsian manifold.

DEFINITION 3.1. An equidistant foliation of M near $\partial_{\infty,+}M$ (resp. $\partial_{\infty,-}M$) is a foliation of a neighborhood of $\partial_{+,\infty}M$ (resp. $\partial_{\infty,-}M$) by locally convex surfaces, $(S_r)_{r\geq r_0}$, for some $r_0 > 0$, such that, for all $r' > r \geq r_0$, $S_{r'}$ is between S_r and $\partial_{\infty,+}M$, and at constant distance r' - r from S_r .

Two equidistant foliations in E will be identified if they coincide in a neighborhood of infinity. In this case they can differ only by the first value r_0 at which they are defined.

Given an equidistant foliation $(S_r)_{r \ge r_0}$ and given $r' > r \ge 0$, there is a natural identification between S_r and $S_{r'}$, obtained by following the normal direction from S_t for all $t \in [r, r']$. This identification will be implicitly used below.

DEFINITION 3.2. Let $(S_r)_{r \geq r_0}$ be an equidistant foliation of M near $\partial_{\infty,+}M$ (resp. $\partial_{\infty,-}M$). The metric at infinity, second and third fundamental forms at infinity associated to $(S_r)_{r \geq r_0}$ are defined by the asymptotic development:

(6)
$$I_r = \frac{1}{2} (e^{2r} I^* + 2II^* + e^{-2r} III^*) ,$$

where I_r is the induced metric on S_r .

Those symmetric 2-tensors I^* , II^* and III^* can naturally be defined on $\partial_{\infty,+}M$ (resp. $\partial_{\infty,-}M$). The existence of the asymptotic development follows from a straightforward computation using the expansion of I_r as a function of r, see [45]. This direct computation shows that, if S_0 exists and is smooth, then

(7)
$$I^* = \frac{1}{2}(I + 2II + III)$$
, $II^* = \frac{1}{2}(I - III)$, $III^* = \frac{1}{2}(I - 2II + III)$,

where I, II and III are the induced metric and second and third fundamental forms of S_0 .

The first part of the following proposition is quite elementary (see e.g. [45]) while the second part follows from ideas of Epstein [24], see below.

PROPOSITION 3.3. The limit metric I^* is in the conformal class at infinity of M.

Let M be a quasifuchsian manifold, and let h be a Riemannian metric on $\partial_{\infty,+}M$ (resp. $\partial_{\infty,-}M$) in the conformal class at infinity of M. There is a unique equidistant foliation near $\partial_{\infty,+}M$ (resp. $\partial_{\infty,-}M$) such that the associated metric at infinity I^{*} is equal to h.

This equidistant foliation can be defined from a metric at infinity in terms of the envelope of a family of horospheres, see [24]. We briefly outline this construction here for completeness. Consider the hyperbolic space \mathbb{H}^3 as the universal cover of M. The metric I^* lifts to a metric on the domain of discontinuity Ω of M, in the canonical conformal class of $\partial_{\infty} \mathbb{H}^3$. Let $x \in \Omega$. For each $y \in \mathbb{H}^3$, the visual metric h_y on $\partial_{\infty}\mathbb{H}^3$ is conformal to I^* . Let $H_{x,r}$ be the set of points $y \in \mathbb{H}^3$ such that $h_y \geq e^{2r}I^*$ at x. A simple computation shows that $H_{x,r}$ is a horosphere intersecting $\partial_{\infty}\mathbb{H}^3$ at x, and the lift of S_r to \mathbb{H}^3 happens to be equal to the boundary of the union of horoballs bounded by the $H_{x,r}$, for $x \in \Omega$.

An alternative approach is provided in [63], in terms of the isometric embedding of the metric h in the "space of horospheres" of \mathbb{H}^3 , and of a duality between this "space of horospheres" and \mathbb{H}^3 .

3.3. Definition and first variation of the W**-volume.** Consider a quasifuchsian manifold M and a geodesically convex subset N of M with smooth boundary. We first define (in Definition 3.4) a modified volume of N, and will then use this modified volume, for a particular choice of a convex subset of M, to define the renormalized volume of M (Definition 3.11).

DEFINITION 3.4. Let $N \subset M$ be a convex subset. We set:

$$W(N) = V(N) - \frac{1}{4} \int_{\partial N} H da$$

where H is the mean curvature of ∂N and da is the area form of its induced metric.

There is a clear similarity between this W-volume and the dual volume of convex subsets of M seen above: only the coefficient changes. The W-volume can thus be considered as the half-sum of the volume and dual volume.

The first variation of this modified volume is computed in [45], using an earlier variation formula for deformations of Einstein manifolds with boundary [61, 60]. Here we consider a first-order deformation of the hyperbolic metric on N (that is, we do not only vary N as a convex subset of M but also allow variations of M), and denote by I' and II', respectively, the corresponding first-order variations of the induced metric and second fundamental form on the boundary of N, and denote the derivatives of all quantities with a prime.

LEMMA 3.5. Under a first-order deformation of N,

(8)
$$W' = \frac{1}{4} \int_{\partial N} \langle II' - \frac{H}{2} I', I \rangle_I da_I \; .$$

PROOF. It was proved in [61, 60] that in this setting the first-order variation of the volume is given by:

$$2V' = \int_{\partial N} H' + \frac{1}{2} \langle I', II \rangle_I da_I \; .$$

The first-order variation of the area form of I is equal to

$$da'_I = \frac{1}{2} \langle I', I \rangle_I da_I$$

and it follows from the definition of W(N) that

$$W' = V' - \frac{1}{4} \left(\int_{\partial N} H da_I \right)' = \int_{\partial N} \frac{H'}{4} + \frac{1}{4} \langle I', II \rangle_I - \frac{1}{8} H \langle I', I \rangle_I da_I .$$

However a simple computation shows that

$$H' = (\langle II, I \rangle_I)' = \langle II', I \rangle_I - \langle II, I' \rangle_I ,$$

and the result follows.

The scalar product appearing in (8) and in the proof between symmetric bilinear forms is the usual extension to tensors of the Riemannian scalar product on $T\partial N$ defined by the induced metric I: if $(e_i)_{i=1,2}$ is any orthonormal basis for I of $T_x\partial N$, and if k, l are two bilinear symmetric forms on $T_x\partial N$, then

$$\langle k,l\rangle_I = \sum_{i,j=1,2} k(e_i,e_j) l(e_i,e_j)$$

at x, and the result does not depend on the choice of $(e_i)_{i=1,2}$.

COROLLARY 3.6. Under the same hypothesis as for Lemma 3.5, we have

(9)
$$W' = \frac{1}{4} \int_{\partial N} H' + \langle II_0, I' \rangle_I da_I ,$$

where $II_0 = II - \frac{H}{2}I$ is the traceless part of II.

The following lemma is a direct consequence of Lemma 3.5.

LEMMA 3.7. Let $r \ge 0$, and let N_r be the set of points of M at distance at most r from N. Then $W(N_r) = W(N) - \pi r \chi(\partial M)$.

PROOF. For $s \in [0, r]$, we denote by N_s the set of points of M at distance at most s from N, and let $w(s) = W(N_s)$. We also denote by I_s, II_s, III_s and B_s the induced metric, second and third fundamental forms and the shape operator of ∂N_s .

According to standard differential geometry formulas, the derivatives of I_s and I_s are given by:

$$I'_s = 2II_s , \quad II'_s = III_s + I_s .$$

Lemma 3.5 therefore shows that:

$$W(N_s)' = \frac{1}{4} \int_{\partial N_s} \langle III_s + I_s - H_sII_s, I_s \rangle da_s = \frac{1}{4} \int_{\partial N_s} \operatorname{tr}(B_s^2) + 2 - H_s^2 da_s =$$
$$= \frac{1}{4} \int_{\partial N_s} 2 - 2 \operatorname{det}(B_s) da_s = \frac{1}{2} \int_{\partial N_s} -K da_s = -\pi \chi(\partial N) . \qquad \Box$$

3.4. First variation of the *W***-volume from infinity.** We have seen above that given a geodesically convex subset $N \subset M$, we have:

- the induced metric I and second fundamental form II on ∂N , as well as the shape operator B defined by the condition that $II = I(B, \cdot) = I(\cdot, B \cdot)$,
- the induced metric I^* and second fundamental form II^* at infinity, as well as the corresponding "shape operator" B^* , defined by $II^* = I^*(B^*\cdot, \cdot) = I^*(\cdot, B^*\cdot)$.

There is a simple expression of I^* and II^* from I and II, and conversely, see (7). One can therefore express the first variation of W in terms of the "data at infinity" I^* and II^* . A key fact, obtained through a lengthy and not very illuminating computation (see [45, Lemma 6.1]) is that Equation (8) remains almost identical when expressed in this manner.

LEMMA 3.8. Under a first-order deformation of N,

(10)
$$W' = -\frac{1}{4} \int_{\partial M} \langle I I^{*\prime} - \frac{H^*}{2} I^{*\prime}, I^* \rangle_{I^*} da_{I^*}$$

Here $H^* = \operatorname{tr}_{I^*} I\!\!I^*$ is the "shape operator at infinity".

COROLLARY 3.9. Under the same hypothesis, we have

(11)
$$W' = -\frac{1}{4} \int_{\partial N} H^{*'} + \langle II_0^*, I^{*'} \rangle_{I^*} da_{I^*} ,$$

where $II_0^* = II^* - \frac{H^*}{2}I^*$ is the traceless part of II^* relative to I^* .

3.5. Definition of the renormalized volume. Consider a Riemannian metric h on ∂M in the conformal class at infinity of $\partial_{\infty} M$. There is by Proposition 3.3 a unique equidistant foliation $(S_r)_{r\geq r_0}$ of M near infinity such that the associated metric at infinity is h.

For $r \ge r_1$, for a fixed $r_1 > 0$, the surfaces S_r bound a convex subset of M, so that Definition 3.4 applies.

DEFINITION 3.10. Let h be a metric on $\partial_{\infty} M$, in the conformal class at infinity. Let $(S_r)_{r \geq r_0}$ be the equidistant foliation close to infinity associated to h. We define $W(M,h) := W(S_r) + \pi r \chi(\partial M)$, for any choice of $r \geq r_1$.

Lemma 3.7 shows that this definition does not depend on the choice of $r \geq r_1$. As a consequence of the definition, for any $\rho \in \mathbb{R}$, $W(M, e^{2\rho}h) = W(M, h) - \pi \rho \chi(\partial M)$.

We can now give the definition of the renormalized volume of M.

DEFINITION 3.11. The renormalized volume V_R of M is defined as equal to W(h) when the metric at infinity h is the unique metric of constant curvature -1 in the conformal class of $\partial_{\infty} M$.

Another possible definition is as the maximum of W(M, h) over all metrics h in the conformal class at infinity of M, under the condition that the area of h is equal to $-2\pi\chi(\partial M)$, see [45]. This is actually an interesting statement: the W-volume can be used to simultaneously uniformize the conformal structures at infinity in the asymptotic boundary components of M.

3.6. The variational formula (3). Consider now a first-order deformation of M, specified — through the Bers Double Uniformization Theorem — by a first-order deformation of the conformal structure at infinity, considered as a point in the Teichmüller space of ∂M .

PROPOSITION 3.12. Under a first-order deformation of the hyperbolic structure on M,

(12)
$$dV_R = -\frac{1}{4} \int_{\partial M} \langle II_0^*, I^{*\prime} \rangle_{I^*} da_{I^*} .$$

Here \langle, \rangle_{I^*} is the extension to symmetric 2-tensors of the Riemannian metric I^* on $T\partial M$. Proposition 3.12 follows by a simple computation from Equation (11), see [45, Lemma 8.5], using the fact that at infinity $H^* = -K^*$ (see [45, Remark 5.4], so that if I^* has constant curvature then II_0^* satisfies the Codazzi equation relative to I^* , as II^* does.

It should be pointed out that Proposition 3.12 has a rather simple translation in terms of complex analysis. Since I_0^* is Codazzi and traceless, it is the real part of a holomorphic quadratic differential, which is minus the Schwarzian derivative q of the uniformization map, see Section 2.3. Moreover, any first-order deformation I^{*} of the hyperbolic metric at infinity determines a first-order variation of the underlying complex structure, and therefore a Beltrami differential μ .

COROLLARY 3.13. Equation (12) can then be written as:

(13)
$$dV_R = -Re\left(\langle q, \mu \rangle\right) = -\int_{\partial M} Re(q\mu) ,$$

where \langle , \rangle is the natural pairing between holomorphic quadratic differentials and Beltrami differentials.

PROOF. The computation needed to go from (12) to (13) is local. We choose a complex coordinate z = x + iy adapted to I^* , that is, such that $I^* = dx^2 + dy^2$ at z = 0. Let $\mu = (\mu_0 + i\mu_1)\frac{dz}{dz}$, and $q = (q_0 + iq_1)dz^2$. A key point is that $I_0^* = Re(q)$, see [45, Appendix A] (note that the sign here is different because of a different convention in the definition of q). Therefore

$$II_0^* = Re(q) = (q_0(dx^2 - dy^2) - 2q_1dxdy) ,$$

while the first-order variation of I^* is equal to

$$I^{*'} = \frac{d}{dt}_{|t=0} |dz(1+t\mu)|^2$$

= $\frac{d}{dt}_{|t=0} |dz+t(\mu_0+i\mu_1)d\bar{z}|^2$
= $2Re((\mu_0+i\mu_1)d\bar{z}^2)$
= $2(\mu_0(dx^2-dy^2)+2\mu_1dxdy)$

As a consequence,

$$\langle II_0^*, I^{*'} \rangle_{I^*} = \langle (q_0(dx^2 - dy^2) - 2q_1dxdy), 2(\mu_0(dx^2 - dy^2) + 2\mu_1dxdy) \rangle_{I^*}$$

= $4(\mu_0q_0 - \mu_1q_1)$,

so that

$$\langle II_0^*, I^* \rangle_{I^*} da_{I^*} = 4Re(q\mu)$$

The result follows by integrating this equality.

3.7. Further properties. The renormalized volume V_R has other properties which can be very interesting, but will not be considered here because they do not (yet) have any analog on the dual volume side. One key property is that when c_- is fixed, the function $V_R(\cdot, c_-) : \mathcal{T}_{\partial+M} \to \mathbb{R}$ is a Kähler potential for the Weil-Petersson metric on $\mathcal{T}_{\partial+M}$. This is proved in [45, Section 8] following ideas from [53].

Another point is that the renormalized volume or closely related quantities are generating functions that can be used to identify symplectic structures with very different definitions on the space of quasifuchsian manifolds [49], or to show that certain maps are symplectic (eg [62], or [46] for the grafting map).

Finally, we already noted that the renormalized volume was originally defined in higher dimensions, in the setting of conformally compact Einstein manifolds [**35**, **31**, **30**]. In this setting, some (but not all, so far) of the properties present in 3 dimensions extend nicely, see [**34**].

4. The extremal length and the measured foliation at infinity

4.1. The measured foliation at infinity. We now focus on the boundary at infinity of quasifuchsian manifolds, and introduce a measured foliation which can be thought of as an analog at infinity of the measured bending lamination on the boundary of the convex core.

DEFINITION 4.1. The measured foliation at infinity is the horizontal measured foliation of q, the Schwarzian derivative of the uniformization map at infinity. We denote it by f.

PROOF OF THEOREM 1.9. According to Lemma 2.2, the extremal length $\operatorname{ext}_{c_{\pm}} f_{\pm}$ is the integral over $\partial_{\infty,\pm} M$ of |q|. By the Nehari estimate (Theorem 5.3), $|q| \leq 3da_{h_{\pm}}/2$, where $da_{h_{\pm}}$ is the area form of the hyperbolic metric h_{\pm} compatible with c_{\pm} . The result follows.

We now consider one connected component of the ideal boundary of M, say $\partial_{\infty,+}M$, equipped with its canonical conformal structure. Recall from Section 2.6 that T_f is the real tree dual to the universal cover of the measured foliation f, and that Φ_f is the Hopf differential of the unique equivariant harmonic map from the universal cover of $\partial_{\infty,+}M$, equipped with this conformal structure, to T_f . The same construction works for $\partial_{\infty,-}M$.

It follows from Theorem 2.3 that $\Phi_f = -q$.

The following lemma relates the renormalized volume to the measured foliation at infinity.

LEMMA 4.2. Let $c \in \mathcal{T}_{\partial M}$, and let $F \in \mathcal{MF}_{\partial M}$. Then F is the measured foliation at infinity of the quasifuchsian hyperbolic metric determined by c if and only if the function Ψ_F defined as

$$\Psi_F = V_R - \frac{1}{4}E_F : \mathcal{T}_{\partial M} \to \mathbb{R}$$

is critical at c.

PROOF. Suppose first that F is the horizontal measured foliation of q, the holomorphic quadratic differential at infinity of the quasifuchsian manifold M(c).

It follows from (5) and (13) that, in a first-order variation \dot{c} ,

$$d\Psi_F(\dot{c}) = dV_R(\dot{c}) - \frac{1}{4}dE_F(\dot{c}) = \operatorname{Re}(\langle q + \Phi_F, \dot{c} \rangle) .$$

But it follows from Theorem 2.3 that $q = -\Phi_F$, and it follows that $d\Psi_F = 0$.

Conversely, if $d\Psi_F = 0$, the same argument as above shows that $q = -\Phi_F$, so that F is the horizontal measured foliation of q.

4.2. Proof of Theorem 1.7. According to Equation (13), in a first-order deformation of M,

$$\dot{V}_R = -\operatorname{Re}(\langle q, \dot{c} \rangle)$$
,

and using Theorem 2.3 we obtain that

$$\dot{V}_R = \operatorname{Re}(\langle \Phi_f, \dot{c} \rangle)$$
.

Using (5), this can be written as

$$\dot{V}_R = -\frac{1}{4} dE_f(\dot{c}) \; .$$

Using Theorem 2.3 again, we finally find that

$$\dot{V}_R = -\frac{1}{2} (d\text{ext}(f))(\dot{c}) \ .$$

5. Comparisons

5.1. Outline. Some applications of the renormalized volume follow from the following related facts, each having its own independent proof.

- (1) The dual volume of the convex core is within a bounded additive constant (depending only on the genus) from the volume of the convex core.
- (2) The dual volume is within a bounded additive constant (depending only on the genus) from the renormalized volume.
- (3) The renormalized volume is bounded from above by the Weil-Petersson distance between the conformal metrics c_{-}, c_{+} on the connected components of its boundary at infinity (times an explicit constant).
- (4) The dual volume is bounded from above by the Weil-Petersson distance between the induced metrics m_{-}, m_{+} on the two boundary components of the convex core (times an explicit function).

We outline the main arguments — and provide references — for those statements below.

Perhaps surprisingly, the comparison between the renormalized volume and the volume of the convex core (suitably corrected by a constant times the length of the measured bending lamination) turns out to be also relevant for applications to physics, see [65].

5.2. Comparing the renormalized volume and the dual volume. The renormalized volume can be compared to the dual volume using the following statement, see [64, Prop. 3.12].

LEMMA 5.1. Let h, h' be two non-positively curved metrics on $\partial_{\infty} M$, in the conformal class at infinity. Suppose that $h' \geq h$ at each point. Then $W(M, h') \geq W(M, h)$, with equality if and only if h = h'.

The proof of this lemma rests on the fact that if $h' \ge h$, then whenever r > 0 is such that the equidistant surfaces S_r and S'_r associated to h and h', respectively, are well-defined (see Section 3.2), then S_r is in the interior of S'_r . And moreover if S_r is in the interior of S'_r , and both S_r and S'_r bound convex subsets, then the W-volume of the domain bounded by S_r is smaller than the W-volume of the domain bounded by $S'_r - W$ is increasing under inclusion of convex subsets.

By definition of the W-volume, we see that W(C(M)) = W(M, h) where h is the metric at infinity defined by the folation of $M \setminus C(M)$ by surfaces equidistant to C(M). A direct computation (see [64]) shows that this metric is equal to $h = h_{Th}/2$, where h_{Th} is Thurston's projective metric. This metric h_{Th} has a simple description when the bending lamination l is supported on simple closed curves: it is then obtained by cutting the induced metric on the boundary of the convex core along the bending curves and inserting for each a flat cylinder of width equal to the exterior bending angle. A key

feature of this metric is that it is in the conformal class at infinity c of M, see [47, 38].

Let h_{-1} be the hyperbolic metric in the conformal class c at infinity. Then

(14)
$$h_{-1} \le h_{Th} \le 2h_{-1}$$
.

The first inequality follows from the definition of the Thurston metric, or from the fact that h_{Th} has curvature at least -1 at all points. The second inequality is a direct consequence of a result of G. Anderson [4, Theorem 4.2], see [13, Theorem 2.1].

It is also useful to remark that if $N \subset M$ is geodesically convex, and if the metric at infinity associated to N is h, then for r > 0 the metric at infinity associated to N_r is $e^{2r}h$. So it follows from Lemma 3.7 that

$$W(M, e^{2r}h) = W(M, h) - \pi r \chi(\partial M)$$

It therefore follows from Equation (14) that:

$$W(M, h_{-1}) \le W(M, h_{Th}) \le W(M, h_{-1}) - \frac{\pi \log(2)}{2} \chi(\partial M)$$
,

so that

$$V_R(M) \le W(C(M)) + \frac{\pi \log(2)}{2} \chi(\partial M) \le V_R(M) - \frac{\pi \log(2)}{2} \chi(\partial M) .$$

Recall that

$$W(C(M)) = V(C(M)) - \frac{1}{4}L_m(l) = V_C^*(M) + \frac{1}{4}L_m(l) ,$$

so we obtain that

$$V_R(M) \le V_C^*(M) + \frac{1}{4}L_m(l) + \frac{\pi \log(2)}{2}\chi(\partial M) \le V_R(M) - \frac{\pi \log(2)}{2}\chi(\partial M)$$
.

Finally, it is known that $L_m(l) \leq 6\pi |\chi(\partial M)|$ (see [13, Theorem 1.1 (2)]), and we therefore obtain the following statement.

THEOREM 5.2. For all quasifuchsian metric on M,

$$V_R(M) - \frac{3\pi |\chi(\partial M)|}{2} + \frac{\pi \log(2)}{2} |\chi(\partial M)| \le V_C^*(M) \le V_R(M) + \pi \log(2) |\chi(\partial M)|.$$

The additive constants depend on the choice of normalization in the definition of renormalized volume — chosing a metric at infinity of constant curvature -2, rather than -1, leads to somewhat simpler additive constants.

Very similar comparison apply when M is a convex co-compact hyperbolic manifold with incompressible boundary, see [14]. For convex cocompact manifolds with compressible boundary, there are also similar estimates [65], but with $V_C^*(M)$ replaced by the W-volume of the convex core, that is, the volume of the convex core minus one *quarter* of the length of the measured bending lamination. **5.3.** An upper bound on the renormalized volume. The renormalized volume of a quasifuchsian manifold can be bounded from above in terms of the Weil-Petersson distance between the conformal metrics on $\partial_{\infty,-}M$ and on $\partial_{\infty,+}M$. This upper bound is based on the following classical result. We denote by D the unit disk in \mathbb{C} , equipped with the hyperbolic metric h.

THEOREM 5.3 (Kraus, Nehari [59]). Let $f: D \to \mathbb{CP}^1$ be an injective holomorphic map. Then at each point $\|\mathcal{S}(f)\|_h \leq 3/2$.

The following theorem from [64] is a direct consequence.

THEOREM 5.4. For any quasifuchsian metric g_0 on $S \times \mathbb{R}$,

(15)
$$V_R(g_0) \le 3\sqrt{\pi(g-1)}d_{WP}(c_-,c_+)$$
,

where c_{-} and c_{+} are the conformal structures at infinity of g_{0} and d_{WP} is the Weil-Petersson distance.

PROOF. Let $c \in \mathcal{T}_S$ be a complex structure on S, let q and μ be a holomorphic quadratic differential and a Beltrami differential on (S, c), and let h' be the first-order variation corresponding to μ of the hyperbolic metric h in the conformal class defined by c. Then a direct computation shows that

$$\int_{S} \langle Re(q), h' \rangle_{h} da_{h} = 4Re\left(\int_{S} q\mu\right)$$

Applying this relation with q equal to Schwarzian derivative term as above, and using that $I_0^* = -Re(q)$, we obtain that for a variation h' of the hyperbolic metric h in the conformal class on the upper component of the boundary at infinity,

$$dV_R(h') = -\frac{1}{4} \int_S \langle II_0^*, h' \rangle_h da_h = \frac{1}{4} \int_S \langle Re(q), h' \rangle_h da_h = Re\left(\int_S q\mu\right) \;.$$

Let z be a local complex coordinate, with $h = \rho^2 |dz|^2$, then we can write

$$q = q' dz^2$$
, $\mu = \mu' \frac{d\bar{z}}{dz}$,

so that

$$dV_R(h') = Re\left(\int_S \left(\frac{q'}{\rho^2}\right) \mu' \rho^2 |dz|^2\right)$$

Using the Nehari estimate (Theorem 5.3) shows that $|q'/\rho^2| \leq 3/2$, and therefore

$$|dV_R(h')| \le \frac{3}{2} \int_S |\mu'| \, \rho^2 |dz|^2 \, .$$

It then follows from the Cauchy-Schwarz inequality that

$$|dV_R(h')| \le \frac{3}{2} \|\mu\|_{WP} \sqrt{4\pi(g-1)} = 3\sqrt{\pi(g-1)} \|\mu\|_{WP}$$
.

We can integrate this inequality on a path from c_{-} to c_{+} as in the first proof above to obtain the result.

5.4. An upper bound on the dual volume. We have seen in Theorem 5.2 that the dual volume $V_C^*(M)$ is within a bounded additive constant from the renormalized volume $V_R(M)$. In addition, Theorem 5.4 shows that $V_R(M)$ is bounded by $3\sqrt{\pi(g-1)}d_{WP}(c_-, c_+)$. This immediately yields the following corollary.

COROLLARY 5.5. For all quasifuchsian metric on
$$M$$
,
(16) $V_C^*(M) \le 3\sqrt{\pi(g-1)}d_{WP}(c_-, c_+) + \pi \log(2)|\chi(\partial M|)$

It should be noted, however, that this argument is quite indirect and uses the whole technology of the renormalized volume.

5.5. Estimates from the dual volume. Recently, Filippo Mazzoli [51] has developed a completely different and much more elementary argument to obtain directly an inequality of the type of (16), and as a consequence an explicit upper bound on $V_C(M)$ in terms of the Weil-Petersson distance between c_- and c_+ , using the dual Bonahon-Schläfli formula.

THEOREM 5.6 (Dual Bonahon-Schläfli formula). Under a first-order variation of a quasifuchsian structure on M,

(17)
$$V_C^{*\prime} = -\frac{1}{2}dL(l)(m') \; .$$

A proof of this formula can be found in [46, Lemma 2.2], based on an analoguous formula proved by Bonahon [8, 9]: under the same hypothesis,

(18)
$$V'_C = \frac{1}{2}L_m(l') \; .$$

Note however that (2) has a much simpler interpretation than (1), since (2) involves only the differential of the (analytic) function L(l) applied to the tangent vector m', while (1) uses the notion of first-order variation of a measured lamination, which is quite subtle and necessitates the full technical toolbox developed by Bonahon [8].

A direct and relatively elementary (but non-trivial) proof of (2) is given by Mazzoli [50], using differential-geometric arguments and an approximation of the boundary of the convex core by smooth surfaces.

Mazzoli then shows [51] that (2) can be used, together with Theorem 1.8, to obtain directly an upper bound on the dual volume.

THEOREM 5.7 (Mazzoli [51]). There exists a constant $K_2 > 0$ such that for any quasifuchsian manifold M and any first-order deformation,

$$|V_C^{*'}| \le K_2 \sqrt{g-1} \|c'\|_{WP}$$
.

It follows directly from this inequality — as in the proof of Theorem 5.4 above — that for any quasifuchsian manifold,

(19)
$$V_C^*(M) \le K_2 \sqrt{g-1} d_{WP}(c_-, c_+) \; .$$

The constant found in [51] is $K_2 = 10.3887$, which is slightly larger than the constant obtained for the renormalized volume in Theorem 5.4.

6. Applications

We briefly outline in this section a few applications of the bound on the renormalized volume, or the dual volume, of quasifuchsian manifolds. This section does not contain complete proofs — we refer to specific papers for the details — but only a very quick outline of the main ideas.

6.1. Bounding the volume of the convex core using the renormalized volume or the dual volume. We have seen in Theorem 5.2 that the renormalized volume $V_R(M)$ is within bounded additive constant (depending only on the genus of the underlying surface) from the dual volume $V_C^*(M)$, while Theorem 5.4 provides an upper bound on the renormalized volume in terms of the Weil-Petersson distance between the conformal metrics at infinity. It follows directly that the volume of the convex core is also bounded in terms of the Weil-Petersson distance between the conformal metrics at infinity: for every genus g > 1, there exists a constant $C_g > 0$ such that for any quasifuchsian manifold M,

$$V_C(M) \le 3\sqrt{\pi(g-1)}d_{WP}(c_-,c_+) + C_g$$
.

It also follows from Theorem 5.7, thanks to the upper bound on the length of the bending lamination in Theorem 1.8, that the same inequality holds for $V_C(M)$, with an additional term $3\pi |\chi(M)|$: for all quasifuchsian manifold M,

$$V_C(M) \le K_2 \sqrt{g-1} d_{WP}(c_-, c_+) + 3\pi |\chi(\partial M)|$$
.

However at this point the constant K_2 arising from Mazzoli's work [51] is somewhat weaker than the $3\sqrt{\pi}$ coming out of the renormalized volume argument.

6.2. The volume of hyperbolic manifolds fibering over the circle. A neat applications of the upper bound found in the previous section on the volume of the convex core is given by Kojima and McShane [41] and Brock and Bromberg [16].

Let $\phi : S \to S$ be a diffeomorphism. The mapping torus of ϕ is the 3-dimensional manifold M_{ϕ} obtained by identifying in $S \times [0, 1]$ the points (x, 1) and $(\phi(x), 0)$ for all $x \in S$. Clearly, M_{ϕ} depends only on the isotopy class of ϕ . Thurston [73] proved that if ϕ is *pseudo-Anosov* (see e.g. [27]) then M_{ϕ} admits a hyperbolic structure, which is unique by the Mostow Rigidity Theorem [58].

A pseudo-Anosov diffeomorphism ϕ acts by pull-back on the Teichmüller space \mathcal{T}_S , and this action is isometric for the Weil-Petersson metric. One can define its Weil-Petersson *translation length*.

$$l(\phi) = \min_{c \in \mathcal{T}_S} d_{WP}(c, \phi^*(c)) \; .$$

Moreover, this minimum is attained along a line, the *axis* of ϕ , on which it acts by translation, see [19].

THEOREM 6.1 (Kojima-McShane, Brock-Bromberg). For any pseudo-Anosov diffeomorphism ϕ of S, $Vol(M_{\phi}) \leq 3\sqrt{\pi(g-1)}l(\phi)$.

The proof of this theorem parallels the construction by Thurston of the hyperbolic structure on M_{ϕ} . Given $c_-, c_+ \in \mathcal{T}_S \times \mathcal{T}_S$, let $M(c_-, c_+)$ be the quasifuchsian hyperbolic structure on $S \times \mathbb{R}$ with conformal structure at infinity c_- and c_+ , respectively. If c is any fixed element of \mathcal{T}_S , Thurston proved that $M(\phi_*^{-n}c, \phi_*^n c) \to \overline{M}$, the infinite cyclic cover of M. Brock and Bromberg [16], building on work of McMullen [52], show that this convergence translates as a precise estimate on the volume of the convex core.

THEOREM 6.2 (Brock-Bromberg [16]). In this setting, $|V_C(\phi_*^{-n}c, \phi_*^n c) - 2nVol(M_{\phi})|$ is bounded.

It follows that

$$V_C(\phi_*^{-n}c,\phi_*^n c) \le 3\sqrt{\pi(g-1)}d_{WP}(\phi_*^{-n}c,\phi_*^n c) + C_g .$$

Taking for c an element of the axis of ϕ , we obtain that

$$V_C(\phi_*^{-n}c,\phi_*^nc) \le 6n\sqrt{\pi(g-1)}l_{\phi} + C_g$$

As a consequence,

$$Vol(M_{\phi}) = \lim_{n \to \infty} \frac{V_C(\phi_*^{-n}c, \phi_*^n c)}{2n} \le 3\sqrt{\pi(g-1)} l_{\phi} ,$$

which is Theorem 6.1

6.3. Systoles of the Weil-Petersson metric on moduli space. As a consequence of Theorem 6.1, Brock and Bromberg obtain a lower bound on the systole \mathcal{M}_S of the moduli space of S, equipped with the Weil-Petersson metric.

COROLLARY 6.3 (Brock-Bromberg). The shortest closed geodesic of the Weil-Petersson metric on \mathcal{M}_S has length at least $Vol(\mathcal{W})/3\sqrt{\pi(g-1)}$.

Here \mathcal{W} is the Weeks manifold, the closed hyperbolic manifold of smallest volume. This statement follows from Theorem 6.1 and from the fact that any closed geodesic of moduli space corresponds to a pseudo-Anosov element of the mapping-class group.

6.4. Entropy and hyperbolic volume of mapping tori. Let ϕ : $S \to S$ be a diffeomorphism. We denote here by $ent(\phi)$ the *entropy* of ϕ , that is, the infimum of the topological entropy of diffeomorphisms isotopic to ϕ . If ϕ is a pseudo-Anosov diffeomorphism, Thurston showed that its entropy is equal to the log of the minimal dilation of diffeomorphisms isotopic to ϕ [27, Exposé 10], and Bers [7] proved that this is equal to its translation length for the Teichmüller distance on \mathcal{T}_S .

Kojima and McShane prove the following relation between the entropy of ϕ and the hyperbolic volume of the mapping torus of ϕ . THEOREM 6.4. If ϕ is a pseudo-Anosov diffeomorphism of S, then

$$\operatorname{ent}(\phi) \ge \frac{1}{3\pi |\chi(S)|} \operatorname{Vol}(M_{\phi})$$
.

The proof of this result is quite similar to the proof of Theorem 6.1 above, but the bound on the renormalized volume of a quasifuchsian manifold by the Weil-Petersson distance between its conformal metrics at infinity is replaced by a bound by the Teichmüller distance between those conformal metrics at infinity. Specifically, Kojima and McShane prove the following statement, see [41, Prop. 11].

THEOREM 6.5 (Kojima–McShane). Let $c_-, c_+ \in \mathcal{T}_S$. Then $V_R(M(c_-, c_+)) \leq 3\pi |\chi(S)| d_T(c_-, c_+)$,

where d_T denotes the Teichmüller distance on \mathcal{T}_S .

The proof is closely related to the proof of Theorem 5.4. We consider a Teichmüller geodesic $(c_t)_{t \in [0,1]}$ with $c_0 = c_-$ and $c_1 = c_+$. Using Corollary 3.13 and the Nehari estimate, Theorem 5.3, we obtain that:

$$V_R(M(c_-, c_+)) \leq \int_0^1 |V_R(c_0, c_t)'| dt$$

$$\leq \int_0^1 \left(\int_{\partial_{\infty, +}M} |\operatorname{Re}(q\mu_t)| \right) dt$$

$$\leq \int_0^1 \left(\int_{\partial_{\infty, +}M} \frac{3}{2} |\mu_t| da_h \right) dt$$

$$\leq 3\pi |\chi(S)| d_T(c_0, c_1) .$$

7. Questions and perspectives

We list here a number of questions concerning the global behavior of the renormalized volume, and in a related way of the measured foliations at infinity, for quasifuchsian manifolds and generalizations.

7.1. Convexity of renormalized volume. Renormalized volume is known to be convex in the neighborhood of the Fuchsian locus, see [57, 18, 74]. However, not much is known on its global behavior. However, it does not appears not to be convex for the Weil-Petersson metric, because, for fixed c_- , there are values of c_+ close to the metric boundary of \mathcal{T}_S for the Weil-Petersson metric where the gradient of $V_R(M(c_-, \cdot))$ is arbitrarily small.²

QUESTION 7.1. Is renormalized volume convex in any other sense, for instance along Teichmüller geodesics or earthquake deformations?

²This argument was communicated by Ken Bromberg.

A related question is whether there is an explicit *lower* bound on the renormalized volume in terms of the Weil-Petersson distance between the conformal metrics at infinity of a quasifuchsian manifold. The existence of such a constant in fact follows from the results of Brock [15] on the volume of the convex core, together with Theorem 5.2, but no estimate of this constant is known.

QUESTION 7.2. Let $g \geq 2$. What is the largest constant $c_g > 0$ for which there exists a constant d > 0 such that, for any quasifuction structure gon $S \times \mathbb{R}$ (where S is a closed surface of genus g) with conformal metrics at infinity $c_{-}, c_{+} \in \mathcal{T}_{S}$,

$$V_R(M,g) \ge c_q d_{WP}(c_-,c_+) - d$$
?

Note that it has been proved recently that the renormalized volume is minimal at the Fuchsian locus (for quasifuchsian manifolds) and for metrics containing a convex core with totally geodesic boundary (for acylindrical manifolds), see [75, 13].

Note that Theorem 5.2 shows that Question 7.2 is equivalent to the corresponding question for the volume or the dual volume of the convex core.

7.2. The measured foliation at infinity. We have described above an analogy between the measured folation at infinity and the measured bending lamination on the boundary of the convex core. This analogy suggests that one may be able to extend to the foliation at infinity a number of known statements or conjectures on the measured bending lamination on the boundary of the convex core. The first question in this direction can be the following.

QUESTION 7.3. Suppose that M is not Fuchsian (that is, it does not contain a closed totally geodesic surface). Do f_{-} and f_{+} fill?

This would be the analog of the well-known (and relatively easy) corresponding statement for l_{-} and l_{+} , the measured bending lamination on the boundary of the convex core.

QUESTION 7.4. Let $(f_-, f_+) \in \mathcal{ML}_S \times \mathcal{ML}_S$, $(f_-, f_+) \neq 0$. Is there at most one quasifuchsian manifold with measured foliation at infinity (f_-, f_+) ?

This is the analog at infinity of the uniqueness part of a conjecture of Thurston on the existence and uniqueness of a quasifuchsian manifold having given measured bending lamination (l_-, l_+) on the boundary of the convex core. In this case (l_-, l_+) are requested to fill and to have no closed leaf of weight larger than π . The existence part of this conjecture for the bending measured lamination was proved in [11], as well as the uniqueness for rational measured laminations, but the uniqueness remains conjectural for more general measured laminations.

A related question would be whether *infinitesimal* rigidity holds, that is, whether any non-zero first-order deformation of M induces a non-zero deformation of either the f_- or f_+ — this might be related to Question 7.1. The analog question for l_- and l_+ is also open.

One can also ask for what pair (f_-, f_+) of measured foliations there *exists* a quasifuchsian manifold having them as measured foliation at infinity:

QUESTION 7.5. Given $(f_-, f_+) \in \mathcal{ML}_S \times \mathcal{ML}_S$, what conditions should it satisfy so that there exists a quasifuchsian manifold M with measured foliation at infinity (f_-, f_+) ?

If the answer to Question 7.3 is positive, then one should ask that (if $(f_-, f_+) \neq 0$) f_- and f_+ should fill. However other conditions might be necessary.

Recently, Diptaishik Choudhury [17] has obtained positive answers to questions 7.3, 7.4 and 7.5 for pairs of *small* foliations, that is, in the neighborhood of the Fuchsian locus. He hows that any pair of filling, small enough measured foliations can be realized as the foliations at infinity of a quasi-fuchsian manifold close to the Fuchsian locus.

7.3. Comparing the foliation at infinity to the measured bending lamination. We have seen above that the renormalized volume of a quasifuchsian manifold is within a bounded additive distance (depending on the genus) from the dual volume of the convex core, and also that the induced metric on the boundary of the convex core is within a bounded quasi-conformal constant of the conformal metric at infinity.

This suggests the following question.

QUESTION 7.6. Is the measured foliation at infinity of a quasifuchsian manifold within bounded distance — in a suitable sense — from the measured bending lamination on the boundary of the convex core?

A recent result of Dumas [21] should be relevant here and actually provides a kind of answer.

7.4. Extension to convex co-compact or geometrically finite hyperbolic manifolds. The definition of renormalized volume can be extended to convex co-compact hyperbolic manifolds, and the main estimates also apply for convex co-compact manifolds with incompressible boundary, see [14]. We can expect Theorem 1.7 to apply to convex co-compact hyperbolic manifolds, and Theorem 1.9 to extend to convex co-compact hyperbolic manifolds with incompressible boundary, while the estimate for manifolds with compressible boundary might involve the injectivity radius of the boundary.

QUESTION 7.7. Can Theorem 1.9 and Theorem 1.7 be extended to geometrically finite hyperbolic 3-manifolds?

Again, the definition and some key properties of renormalized volume extend to geometrically finite hyperbolic 3-manifolds, see [33]. It could be expected that Theorems 1.7 and 1.9 extends to this setting. (For Theorem 1.9 one might need consider manifolds with incompressible boundary.)

7.5. Higher dimensions. Finally the notion of renormalized volume can be considered in dimension larger than 3, and a number of questions appear to be meaningful.

QUESTION 7.8. Are there any extensions of the measured foliation at infinity in higher dimension, for quasifuchsian (or convex co-compact) hyperbolic *d*-dimensional manifolds?

For those manifolds, there is a well-defined notion of convex core, and the boundary of the convex core also has a "pleating". However the pleating lamination might have a more complex structure than for d = 3, with codimension 1 "pleating hypersurfaces" of the boundary meeting along singular strata of higher codimension. Other aspects of the renormalized volume of quasifuchsian manifold have a partial extension in higher dimensions, see e.g. [34].

References

- L. V. Ahlfors. Lectures on quasiconformal mappings. D. Van Nostrand Co., Inc., Toronto, Ont.-New York-London, 1966. Manuscript prepared with the assistance of Clifford J. Earle, Jr. Van Nostrand Mathematical Studies, No. 10. MR 0200442
- [2] Lars V. Ahlfors. Some remarks on Teichmüller's space of Riemann surfaces. Ann. of Math. (2), 74:171–191, 1961. MR 0204641
- [3] Lars V. Ahlfors. Curvature properties of Teichmüller's space. J. Analyse Math., 9:161–176, 1961/1962. MR 0136730
- [4] Charles Gregory Anderson. Projective structures on Riemann surfaces and developing maps to H3 and CPn. University of California, Berkeley, 1998. MR 2698860
- [5] Lars Andersson, Thierry Barbot, Riccardo Benedetti, Francesco Bonsante, William M. Goldman, François Labourie, Kevin P. Scannell, and Jean-Marc Schlenker. Notes on: "Lorentz spacetimes of constant curvature" [Geom. Dedicata 126 (2007), 3–45; MR2328921] by G. Mess. *Geom. Dedicata*, 126:47–70, 2007. MR 2328922
- [6] Lipman Bers. Simultaneous uniformization. Bull. Amer. Math. Soc., 66:94–97, 1960. MR 0111834
- [7] Lipman Bers. An extremal problem for quasiconformal mappings and a theorem by Thurston. Acta Math., 141(1-2):73–98, 1978. MR 0477161
- [8] Francis Bonahon. A Schläfli-type formula for convex cores of hyperbolic 3-manifolds. J. Differential Geom., 50(1):25–58, 1998. MR 1678473
- [9] Francis Bonahon. Variations of the boundary geometry of 3-dimensional hyperbolic convex cores. J. Differential Geom., 50(1):1–24, 1998. MR 1678469
- [10] Francis Bonahon. Geodesic laminations on surfaces. In Laminations and foliations in dynamics, geometry and topology (Stony Brook, NY, 1998), volume 269 of Contemp. Math., pages 1–37. Amer. Math. Soc., Providence, RI, 2001. MR 1810534
- [11] Francis Bonahon and Jean-Pierre Otal. Laminations mesurées de plissage des variétés hyperboliques de dimension 3. Ann. Math., 160:1013–1055, 2004. MR 2144972
- [12] Martin Bridgeman. Average bending of convex pleated planes in hyperbolic threespace. Invent. Math., 132(2):381–391, 1998. MR 1621436
- [13] Martin Bridgeman, Jeffrey Brock, and Kenneth Bromberg. Schwarzian derivatives, projective structures, and the Weil-Petersson gradient flow for renormalized volume. *Duke Math. J.*, 168(5):867–896, 2019. MR 3934591
- [14] Martin Bridgeman and Richard D. Canary. Renormalized volume and the volume of the convex core. Ann. Inst. Fourier (Grenoble), 67(5):2083–2098, 2017. MR 3732685

- [15] Jeffrey F. Brock. The Weil-Petersson metric and volumes of 3-dimensional hyperbolic convex cores. J. Amer. Math. Soc., 16(3):495–535 (electronic), 2003. MR 1969203
- [16] Jeffrey F. Brock and Kenneth W. Bromberg. Inflexibility, Weil-Peterson distance, and volumes of fibered 3-manifolds. *Math. Res. Lett.*, 23(3):649–674, 2016. MR 3533189
- [17] Diptaishik Choudhury. Measured foliations at infinity of quasi-fuchsian manifolds near the fuchsian locus. arXiv preprint arXiv:2111.01614, 2021.
- [18] Corina Ciobotaru and Sergiu Moroianu. Positivity of the renormalized volume of almost-Fuchsian hyperbolic 3-manifolds. Proc. Amer. Math. Soc., 144(1):151–159, 2016. MR 3415585
- [19] Georgios Daskalopoulos and Richard Wentworth. Classification of Weil-Petersson isometries. Amer. J. Math., 125(4):941–975, 2003. MR 1993745
- [20] Georgios D. Daskalopoulos and Richard A. Wentworth. Harmonic maps and Teichmüller theory. In Handbook of Teichmüller theory. Vol. I, volume 11 of IRMA Lect. Math. Theor. Phys., pages 33–109. Eur. Math. Soc., Zürich, 2007. MR 2349668
- [21] David Dumas. The Schwarzian derivative and measured laminations on Riemann surfaces. Duke Math. J., 140(2):203–243, 2007. MR 2359819
- [22] David Dumas. Complex projective structures. In Handbook of Teichmüller theory. Vol. II, volume 13 of IRMA Lect. Math. Theor. Phys., pages 455–508. Eur. Math. Soc., Zürich, 2008. MR 2497780
- [23] Charles L. Epstein. The hyperbolic Gauss map and quasiconformal reflections. J. Reine Angew. Math., 372:96–135, 1986. MR 0863521
- [24] Charles L. Epstein. The hyperbolic Gauss map and quasiconformal reflections. J. Reine Angew. Math., 372:96–135, 1986. MR 0863521
- [25] D. B. A. Epstein and A. Marden. Convex hulls in hyperbolic spaces, a theorem of Sullivan, and measured pleated surfaces. In D. B. A. Epstein, editor, Analytical and geometric aspects of hyperbolic space, volume 111 of L.M.S. Lecture Note Series. Cambridge University Press, 1986. MR 0903852
- [26] D. B. A. Epstein and V. Markovic. The logarithmic spiral: a counterexample to the K = 2 conjecture. Ann. of Math. (2), 161(2):925–957, 2005. MR 2153403
- [27] A. Fathi, F. Laudenbach, and V. Poenaru. Travaux de Thurston sur les surfaces. Société Mathématique de France, Paris, 1991. Séminaire Orsay, Reprint of Travaux de Thurston sur les surfaces, Soc. Math. France, Paris, 1979 [MR 82m:57003], Astérisque No. 66-67 (1991). MR 0572418
- [28] A. E. Fischer and A. J. Tromba. On the Weil-Petersson metric on Teichmüller space. *Trans. Amer. Math. Soc.*, 284(1):319–335, 1984. MR 0742427
- [29] William M. Goldman. Geometric structures on manifolds and varieties of representations. In Geometry of group representations (Boulder, CO, 1987), volume 74 of Contemp. Math., pages 169–198. Amer. Math. Soc., Providence, RI, 1988. MR 0957518
- [30] C. Robin Graham. Volume and area renormalizations for conformally compact Einstein metrics. In *The Proceedings of the 19th Winter School "Geometry and Physics"* (Srní, 1999), number 63, pages 31–42, 2000. MR 1758076
- [31] C. Robin Graham and Edward Witten. Conformal anomaly of submanifold observables in AdS/CFT correspondence. *Nuclear Phys. B*, 546(1-2):52–64, 1999. MR 1682674
- [32] Mikhail Gromov and Richard Schoen. Harmonic maps into singular spaces and p-adic superrigidity for lattices in groups of rank one. Inst. Hautes Études Sci. Publ. Math., (76):165–246, 1992. MR 1215595
- [33] Colin Guillarmou, Sergiu Moroianu, and Frédéric Rochon. Renormalized volume on the Teichmüller space of punctured surfaces. Ann. Sc. Norm. Super. Pisa Cl. Sci. (5), 17(1):323–384, 2017. MR 3676051
- [34] Colin Guillarmou, Sergiu Moroianu, and Jean-Marc Schlenker. The renormalized volume and uniformization of conformal structures. J. Inst. Math. Jussieu, 17(4):853–912, 2018. MR 3835525

- [35] Mark Henningson and Kostas Skenderis. The holographic weyl anomaly. JHEP, 9807:023, 1998. MR 1644988
- [36] Craig D. Hodgson and Igor Rivin. A characterization of compact convex polyhedra in hyperbolic 3-space. *Invent. Math.*, 111:77–111, 1993. MR 1193599
- [37] John Hubbard and Howard Masur. Quadratic differentials and foliations. Acta Math., 142(3-4):221–274, 1979. MR 0523212
- [38] Yoshinobu Kamishima and Ser P. Tan. Deformation spaces on geometric structures. In Aspects of low-dimensional manifolds, volume 20 of Adv. Stud. Pure Math., pages 263–299. Kinokuniya, Tokyo, 1992. MR 1208313
- [39] Steven P. Kerckhoff. The asymptotic geometry of Teichmüller space. *Topology*, 19(1):23–41, 1980. MR 0559474
- [40] Steven P. Kerckhoff. Earthquakes are analytic. Comment. Math. Helv., 60(1):17–30, 1985. MR 0787659
- [41] Sadayoshi Kojima and Greg McShane. Normalized entropy versus volume for pseudo-Anosovs. Geom. Topol., 22(4):2403–2426, 2018. MR 3784525
- [42] Nicholas J. Korevaar and Richard M. Schoen. Sobolev spaces and harmonic maps for metric space targets. Comm. Anal. Geom., 1(3-4):561–659, 1993. MR 1266480
- [43] Nicholas J. Korevaar and Richard M. Schoen. Global existence theorems for harmonic maps to non-locally compact spaces. *Comm. Anal. Geom.*, 5(2):333–387, 1997. MR 1483983
- [44] Kirill Krasnov. Holography and Riemann surfaces. Adv. Theor. Math. Phys., 4(4):929–979, 2000. MR 1867510
- [45] Kirill Krasnov and Jean-Marc Schlenker. On the renormalized volume of hyperbolic 3-manifolds. Comm. Math. Phys., 279(3):637–668, 2008. MR 2386723
- [46] Kirill Krasnov and Jean-Marc Schlenker. A symplectic map between hyperbolic and complex Teichmüller theory. Duke Math. J., 150(2):331–356, 2009. MR 2569616
- [47] Ravi S. Kulkarni and Ulrich Pinkall. A canonical metric for Möbius structures and its applications. Math. Z., 216(1):89–129, 1994. MR 1273468
- [48] François Labourie. Métriques prescrites sur le bord des variétés hyperboliques de dimension 3. J. Differential Geom., 35:609–626, 1992. MR 1163450
- [49] Brice Loustau. Minimal surfaces and symplectic structures of moduli spaces. Geom. Dedicata, 175:309–322, 2015. MR 3323643
- [50] Filippo Mazzoli. The dual Bonahon-Schläfli formula. Algebr. Geom. Topol., 21(1):279–315, 2021. MR 4224742
- [51] Filippo Mazzoli. The dual volume of quasi-Fuchsian manifolds and the Weil-Petersson distance. Trans. Amer. Math. Soc., 375(1):695–723, 2022. MR 4358680
- [52] Curtis T. McMullen. Renormalization and 3-manifolds which fiber over the circle, volume 142 of Annals of Mathematics Studies. Princeton University Press, Princeton, NJ, 1996. MR 1401347
- [53] Curtis T. McMullen. The moduli space of Riemann surfaces is Kähler hyperbolic. Ann. of Math. (2), 151(1):327–357, 2000. MR 1745010
- [54] Geoffrey Mess. Lorentz spacetimes of constant curvature. Geom. Dedicata, 126:3–45, 2007. MR 2328921
- [55] John Milnor. The Schläfli differential equality. In *Collected papers, vol. 1.* Publish or Perish, 1994. MR 2307956
- [56] John W. Morgan and Peter B. Shalen. Valuations, trees, and degenerations of hyperbolic structures. I. Ann. of Math. (2), 120(3):401–476, 1984. MR 0769158
- [57] Sergiu Moroianu. Convexity of the renormalized volume of hyperbolic 3-manifolds. Amer. J. Math., 139(5):1379–1394, 2017. MR 3702501
- [58] G. D. Mostow. Quasi-conformal mappings in n-space and the rigidity of hyperbolic space forms. Inst. Hautes Études Sci. Publ. Math., (34):53–104, 1968. MR 0236383
- [59] Zeev Nehari. The Schwarzian derivative and schlicht functions. Bull. Amer. Math. Soc., 55:545–551, 1949. MR 0029999

- [60] Igor Rivin and Jean-Marc Schlenker. The Schläfli formula in Einstein manifolds with boundary. *Electronic Research Announcements of the A.M.S.*, 5:18–23, 1999. MR 1669399
- [61] Igor Rivin and Jean-Marc Schlenker. The Schläfli formula and Einstein manifolds. Preprint math.DG/0001176, 2000.
- [62] Carlos Scarinci and Jean-Marc Schlenker. Symplectic Wick rotations between moduli spaces of 3-manifolds. Ann. Sc. Norm. Super. Pisa Cl. Sci. (5), 18(3):781–829, 2018. MR 3807587
- [63] Jean-Marc Schlenker. Hypersurfaces in Hⁿ and the space of its horospheres. Geom. Funct. Anal., 12(2):395–435, 2002. MR 1911666
- [64] Jean-Marc Schlenker. The renormalized volume and the volume of the convex core of quasifuchsian manifolds. *Math. Res. Lett.*, 20(4):773–786, 2013. Corrected version available as arXiv:1109.6663v4. MR 3188032
- [65] Jean-Marc Schlenker and Edward Witten. No Ensemble Averaging Below the Black Hole Threshold. arXiv e-prints, arXiv:2202.01372, February 2022. Now accepted for publication in J. of High Energy Physics.
- [66] Richard Schoen. The effect of curvature on the behavior of harmonic functions and mappings. In Nonlinear partial differential equations in differential geometry (Park City, UT, 1992), volume 2 of IAS/Park City Math. Ser., pages 127–184. Amer. Math. Soc., Providence, RI, 1996. MR 1369588
- [67] Dennis Sullivan. Travaux de Thurston sur les groupes quasi-fuchsiens et les variétés hyperboliques de dimension 3 fibrées sur S¹. In Bourbaki Seminar, Vol. 1979/80, volume 842 of Lecture Notes in Math., pages 196–214. Springer, Berlin-New York, 1981. MR 0636524
- [68] L. Takhtajan and P. Zograf. On uniformization of Riemann surfaces and the Weil-Petersson metric on the Teichmüller and Schottky spaces. *Mat. Sb.*, 132:303–320, 1987. English translation in *Math. USSR Sb. 60:297-313, 1988.* MR 0889594
- [69] Leon Takhtajan and Peter Zograf. Hyperbolic 2-spheres with conical singularities, accessory parameters and Kähler metrics on $M_{0,n}$. Trans. Amer. Math. Soc., 355(5):1857–1867 (electronic), 2003. MR 1953529
- [70] Leon A. Takhtajan and Lee-Peng Teo. Liouville action and Weil-Petersson metric on deformation spaces, global Kleinian reciprocity and holography. *Comm. Math. Phys.*, 239(1-2):183–240, 2003. MR 1997440
- [71] William P. Thurston. Three-dimensional geometry and topology. Originally notes of lectures at Princeton University, 1979. Recent version available on http://www.msri.org/publications/books/gt3m/, 1980.
- [72] William P. Thurston. Zippers and univalent functions. In *The Bieberbach conjecture* (West Lafayette, Ind., 1985), volume 21 of Math. Surveys Monogr., pages 185–197.
 Amer. Math. Soc., Providence, RI, 1986. MR 0875241
- [73] William P Thurston. Hyperbolic structures on 3-manifolds, ii: Surface groups and 3-manifolds which fiber over the circle. arXiv preprint math/9801045, 1998.
- [74] Franco Vargas Pallete. Local convexity of renormalized volume for rank-1 cusped manifolds. ArXiv e-prints, May 2015. MR 4028105
- [75] Franco Vargas Pallete. Continuity of the renormalized volume under geometric limits. ArXiv e-prints, May 2016.
- [76] André Weil. Modules des surfaces de Riemann. In Séminaire Bourbaki; 10e année: 1957/1958. Textes des conférences; Exposés 152à 168; 2e éd.corrigée, Exposé 168, page 7. Secrétariat mathématique, Paris, 1958. MR 0124485
- [77] Richard A. Wentworth. Energy of harmonic maps and Gardiner's formula. In In the tradition of Ahlfors-Bers. IV, volume 432 of Contemp. Math., pages 221–229. Amer. Math. Soc., Providence, RI, 2007. MR 2342819
- [78] Michael Wolf. Harmonic maps from surfaces to R-trees. Math. Z., 218(4):577–593, 1995. MR 1326987

- [79] Michael Wolf. On the existence of Jenkins-Strebel differentials using harmonic maps from surfaces to graphs. Ann. Acad. Sci. Fenn. Ser. A I Math., 20(2):269–278, 1995. MR 1346812
- [80] Michael Wolf. On realizing measured foliations via quadratic differentials of harmonic maps to R-trees. J. Anal. Math., 68:107–120, 1996. MR 1403253
- [81] Scott Wolpert. Noncompleteness of the Weil-Petersson metric for Teichmüller space. Pacific J. Math., 61(2):573–577, 1975. MR 0422692
- [82] Scott A. Wolpert. Geometry of the Weil-Petersson completion of Teichmüller space. In Surveys in differential geometry, Vol. VIII (Boston, MA, 2002), Surv. Differ. Geom., VIII, pages 357–393. Int. Press, Somerville, MA, 2003. MR 2039996

DEPARTMENT OF MATHEMATICS, UNIVERSITY OF LUXEMBOURG, MAISON DU NOM-BRE, 6 AVENUE DE LA FONTE, L-4364 ESCH-SUR-ALZETTE, LUXEMBOURG,

Email address: jean-marc.schlenker@uni.lu